%% Template for writing an Oberwolfach Report
%% maintained by Oliver Wienand <reports at mfo dot de>
%% last change on October 18th, 2005

%% Before submitting the report to <reports at mfo dot de>
%% you can run automated tests for common errors online
%% on the websites of the institute.
%% For further informations and the online test visit
%% http://www.mfo.de/cgi-bin/path?publications/owr/diagnostics.html

\documentclass[report]{owrart}



        \usepackage{color}
\newcommand{\red}[1]{\textcolor{red}{\bf #1}}


\begin{document}
\begin{thebibliography}{99}

\bibitem{Ewald-Ishida}
G.~Ewald and M-N Ishida, \emph{Completion of real fans and Zariski-Riemann spaces},
 Toh\^oku Math. J., (2), 58, (2006), 189-218.
 
  \bibitem{GKZ}
I.~M. Gel'fand, M.~M. Kapranov, and A.~V. Zelevinsky,
  \emph{Discriminants, resultants, and multidimensional determinants},
  Mathematics: Theory \& Applications, Birkh\"auser Boston Inc., Boston, MA,
  1994.

  \bibitem{GP-T2}
P.~D. Gonz\'alez~P\'erez and B.~Teissier, \emph{Toric geometry and the Semple-Nash modification}, submitted. See also arXiv:0912.0593

\bibitem{GS-CRAS}
G.~Gonzalez~Sprinberg, \emph{Transform\'e de {N}ash et \'eventail de dimension {$2$}}, C. R.
  Acad. Sci. Paris S\'er. A-B \textbf{284} (1977), no.~1, A69--A71.
 
 \bibitem{No}
A.~Nobile, \emph{Some properties of the {N}ash blowing-up}, Pacific J. Math.
  \textbf{60} (1975), no.~1, 297--305.
  
  \bibitem{S}
J.~G. Semple, \emph{Some investigations in the geometry of curve and surface elements}, Proc. London Math. Soc. (3) \textbf{4} (1954), 24--49.

\bibitem{Spivakovsky}
M.~Spivakovsky, \emph{Sandwiched singularities and desingularization of
  surfaces by normalized Nash transformations}, Ann. of Math. (2)
  \textbf{131} (1990), no. 3, 411--491.
  
 \bibitem{Sumihiro-I}
H.~Sumihiro, \emph{Equivariant completion}, J. Math. Kyoto Univ. \textbf{14}
  (1974), 1--28.

 
 \end{thebibliography}

\reporter{Lars Kastner}
%% -------------------------------------------------------------------------------
%% Enter additionally required packages below this comment.
%% Please be conservative and only require packages
%% which are commonly available. Do not use any packages,
%% which alter the page layout (such as 'a4', etc.)
%%
%% For the inclusion of graphics, .eps files and the use
%% of the psfig, epsfig, graphics or graphicx packages
%% are preferred.
%% Please avoid including pixel-based graphics formats
%% (such as .gif, .png, .jpg) and the pdf format.
%% -------------------------------------------------------------------------------

%% -------------------------------------------------------------------------------
%% VARS: Report-specific settings.
%% -------------------------------------------------------------------------------





\renewcommand{\qedsymbol}{$\square$}
\def \findem{\smallskip\hfill\vrule width 5 pt height 7 pt depth - 2 pt\smallskip}
\def\Par{{\par  \noindent}}
\def\A{{\mathbf A}}
\def\Aa{{\mathcal A}}
\def\P{{\mathbf P}}
\def\Pp{{\mathcal P}}
\def\Q{{\mathbf Q}}
\def\Qq{{\mathcal Q}}
\def\Rr{{\mathcal R}}
\def\R{{\mathbf R}}
\def\B{{\mathcal B}}
\def\Ss{{\mathcal S}}
\def\C{{\mathbf C}}
\def\L{{\mathcal L}}
\def\N{{\mathbf N}}
\def\Newton{{\mathcal N}}
\def\T{{\mathbf T}}
\def\Z{{\mathbf Z}}
\def\O{{\mathcal O}}
\def\og{{\omega}}
\def\G{{\Gamma}}





\begin{report}

%% -------------------------------------------------------------------------------
%% First example abstract.
%%
%% Please use the environment talk, it has three obligatory and one
%% optional argument. The syntax is:
%% -----------------------
%% \begin{talk}[coauthors]{Name of the speaker}{Title of the talk}{Author Sorting Index}
%%      .....
%% \end{talk}
%% -----------------------
%% The names of coauthors will appear in form of :
%% (joint work with ...)
%%
%% The Author Sorting Index is only used for sorting the list of all authors
%% by their last name. It should not contain special characters like
%% accents, german umlaute, etc. (just replace \"a by a, \'a by a, etc.)
%%
%% Please use the standard thebibliography environment to include
%% your references, and try to use labels for the bibitems, which
%% are uniquely assigned to you in order to avoid conflicts with other authors.
%% -------------------------------------------------------------------------------

\begin{talk}[Pedro Daniel Gonz\'alez P\'erez]{Bernard Teissier}
{Toric geometry and the Semple-Nash modification}
{Teissier, Bernard}

\noindent
The Semple-Nash modification of an algebraic variety over a field of characteristic zero is a birational map $\nu\colon NX\to X$ such that $\nu^*\Omega^1_X$ has a locally free quotient of rank $d={\rm dim}X$ and which is minimal for this property. It is unique up to unique $X$-isomorphism. Geometrically the set-theoretic fiber $\vert \nu^{-1}(x)\vert$ is the set of limit positions at $x$ of tangent spaces to $X$ at non singular points tending to $x$. In \cite{S} Semple asks whether iterating this operation leads to a non singular model of $X$. The same question was asked by John Nash in the late 1960's. Nobile proved in \cite{No} that the map $\nu$ is an isomorphism if and only if $X$ is non singular, so that to answer positively Semple's question it suffices to prove that for some $k$ we have $N^{k+1}X=N^kX$. The case of dimension one is easily settled positively, and G\'erard Gonzalez~Sprinberg proved in \cite{GS-CRAS} that iterating the operation of
\emph{normalized} Semple-Nash modification (i.e., at each step, the modification followed by normalization) eventually resolves singularities of 2-dimensional normal toric varieties. This eventually led to the best result on this problem to this day, which is due to Spivakovsky (see \cite{Spivakovsky}): iterating the normalized Semple-Nash modification eventually resolves singularities of normal surfaces over an algebraically closed field.\par
Our basic idea is that in higher dimensions it should be easier in the toric case to deal with the Semple-Nash modification without normalization. Gonzalez Sprinberg showed that 
the Semple-Nash modification, in characteristic zero, is isomorphic with the blowing up of the
\emph{logarithmic jacobian ideal}, which is an equivariant sheaf of ideals defined as follows:\par
Let $k$ be a field and let $\Gamma$ be a finitely generated subgroup of a lattice $M\simeq\Z^d$. Denote by $(\gamma_i)_{i\in F}$ a set of generators. The logarithmic jacobian ideal is the monomial ideal of the semigroup algebra $k[t^\Gamma]$ generated by the elements
$t^{\gamma_{i_1}+\cdots+\gamma_{i_d}}$ for all $(i_1,\ldots,i_d)\subset F$ such that $\gamma_{i_1}\wedge\ldots\wedge\gamma_{i_d}\neq 0\ {\rm in}\ M_\R$.

Since we wanted to describe precisely blowing-ups (and not only normalized blowing-ups) and modifications we wrote down the basic theory of toric varieties without any assumption of normality or of projectivity. However Sumihiro's
theorem (see \cite{Sumihiro-I}) on the existence of a covering by
invariant affine varieties of a variety on which a torus of the same dimension acts with a dense orbit fails without the assumption of normality, and we have to set the existence of such a covering as part
of the definition of a toric variety. Then an abstract toric
variety has a combinatorial description: it corresponds to certain
semigroups in the convex duals of the cones of a fan, which
satisfy a natural gluing-up condition. This generalizes the
definition of \cite{GKZ} which concerns toric varieties
equivariantly embedded in projective space.\par

Given a rational strictly convex cone $\sigma\in N_\R$ and a finitely
generated semigroup $\Gamma$, such that  $\Z \Gamma = M$ and $\R_{\geq 0 } \Gamma = 
\check\sigma\subset M_\R$ 
 one has the affine toric variety  ${\rm Spec}k[t^\Gamma]$ 
 with its torus $T^M ={\rm Spec}k[t^M] $.
For a face $\tau<\sigma$, the lattice $M(\tau;\Gamma)$
spanned by $\Gamma\cap\tau^\perp$ is in general a sublattice of finite index 
in $M(\tau)=M\cap\tau^\perp$.  
This corresponds to the fact that
 the normalization map is equivariant and its restriction to 
 an orbit of the normalization ${\rm Spec}k[t^{\check\sigma\cap M}]$ 
 may not be one to one.      
The variety $T^{\Gamma\cap\tau^\perp}={\rm Spec}k[t^{\Gamma\cap\tau^\perp}]$
with torus $T^{M(\tau;\Gamma)}$  is an orbit closure in  ${\rm Spec}k[t^\Gamma]$.
\par
The semigroup $\Gamma_\tau= \Gamma+M(\tau;\Gamma)$ is again a finitely generated semigroup, generating the group $M$ and with the property that the cone $\R_{\geq 0}\Gamma_\tau$ which it generates in $M_\R$ is equal to $\check\tau$.\par
A toric variety is a triple $(N,\Sigma, \mathbf{\Gamma})$ consisting of a lattice $N$, a fan $\Sigma$ in $N_\R$ and a family of subsemigroups of the dual lattice $M$ of $N$, $$ \mathbf{\Gamma}=(\Gamma_\sigma)_{\sigma\in \Sigma}$$ such that:\par\noindent
$\bullet$ The group $\Z\Gamma_\sigma$ generated by $\Gamma_\sigma$ is equal to $M$ for all $\sigma\in\Sigma$.\par\noindent
$\bullet$ The cone $\R_{\geq 0}\Gamma_\sigma$ generated by $\Gamma_\sigma$ is equal to $\check\sigma$ for all $\sigma\in\Sigma$.\par\noindent
$\bullet$ Far all $\sigma\in\Sigma$ and $\tau<\sigma$, we have $\Gamma_\tau= \Gamma_\sigma+M(\tau;\Gamma_\sigma)$.\par
The toric variety $T_\Sigma^ \mathbf{\Gamma}=\bigcup_{\sigma\in\Sigma}T^{\Gamma_\sigma}$ is then obtained by gluing up the affine toric varieties $T^{\Gamma_\sigma}$ as in the normal case.\par
Now if we define an abstract toric variety as an irreducible separated algebraic variety $X$ over $k$ equipped with an algebraic action of a torus $T^M\subset X$ of the same dimension extending the action of the torus on itself by translation and covered by finitely many open affine invariant subsets, then we have:\par\medskip\noindent
\textbf{Theorem 1}. (See \cite{GP-T2})
\emph{An abstract toric variety is 
equivariantly isomorphic to a variety $T_\Sigma^ \mathbf{\Gamma}$.}\par
In fact, with the natural definitions of toric morphisms, this sets up an equivalence of categories, see \cite{GP-T2} for details.\par\noindent
The logarithmic jacobian ideals sheafify into an equivariant sheaf of ideals on $T_\Sigma^ \mathbf{\Gamma}$.\par\smallskip Given a monomial ideal $I=(t^{n_1},\ldots ,t^{n_s})$ in $k[t^\Gamma]$, it determines a piecewise linear function ${\rm ord}_I\colon \sigma\to \R$ by $\nu\mapsto{\rm min}_{m\in I}\langle\nu,m\rangle$. The cones of linearity $\sigma_i$ correspond to the vertices $(m_1,\ldots ,m_k)$ in $\check\sigma$ of the Newton polyhedron of $I$ and they are the cones of maximal dimension of a fan $\Sigma_I$ with support $\vert\sigma\vert$. If to each cone $\sigma_i\subset\sigma$ we attach the semigroup $\Gamma_i=\Gamma+\langle (n_j-m_i)_{j=1,\ldots, s}\rangle\subset \check\sigma_i\cap M$ and attach to the faces of the $\sigma_i$ the semigroups corresponding to the gluing up rules explained above, we see that we have a triple $(N,\Sigma_I,\mathbf{\Gamma}_I)$ corresponding to a toric variety $T_{\Sigma_I}^ {\mathbf{\Gamma}_I}$ endowed with an equivariant proper map to $T^\Gamma$. It is the blowing-up of the ideal $I$. This construction globalizes into the blowing up of a sheaf of equivariant ideals on a toric variety, and we can apply it to the sheaf of logarithmic jacobian ideals to obtain an analogue of the Semple-Nash modification, which is defined over any field.\par 
The analogue in the toric case, in any characteristic, of Nobile's result is an easy combinatorial lemma: the blowing up of the logarithmic jacobian ideal is an isomorphism if and only if the toric variety $T_\Sigma^ \mathbf{\Gamma}$ is non singular.\par\noindent
So iterating the logarithmic jacobian blowing up of a toric variety $T_\Sigma^ \mathbf{\Gamma}$ produces a sequence of refinements $\Sigma^{(j)}$ of the fan $\Sigma$, with attached systems of semigroups $\mathbf{\Gamma}^{(j)}$. The goal is to prove that they stabilize: the refinement stops only when the space is non singular, as we have seen.\par
Ewald and Ishida have introduced in \cite{Ewald-Ishida} the analogue in toric geometry of the Zariski-Riemann manifold in algebraic geometry: it is the space $ZR(M)$ of (additive) preorderings $\leq$ on $M$ in the following sense:\par\noindent
 \begin{itemize}
 \item $\forall \ m, n\in M,\ m\leq n\ {\rm or}\ n\leq m$.
 \item $m\leq n\ {\rm and}\ n\leq p$ imply $m\leq p$.
 \item $m\leq n$ implies $m+p\leq n+p,\ \forall p\in M$,
 \end{itemize}
 endowed with the topology defined by the basis of open sets $${\mathcal U}(\theta)=\{\nu\in ZR(M)/\check\theta\subset L(\nu)\},$$ where $\theta$ is a rational convex cone of $N_\R$ and $L(\nu)$ is the semigroup of elements of $M$ which are $\geq_\nu0$, or non negative for the preorder $\nu$.\par
 With this topology, the space $ZR(M)$ is quasi compact, and it behaves like a space of valuations. In particular the preorders defined by vectors $\nu$ of the dual lattice $N$ of $M$, as $m\leq_\nu n\iff \langle m,\nu\rangle\leq \langle m,\nu\rangle$ correspond to divisorial valuations. A preorder $\nu\in ZR(M)$ picks a cone in each refinement $\Sigma^{(j)}$ of $\Sigma$, and to prove that the sequence of toric varieties $(N,\Sigma^{(j)},\mathbf{\Gamma}^{(j)})$ stabilizes, it suffices to check that for every preorder it stabilizes along the sequence of cones picked by that preorder. Then we have the following partial result:\par\medskip\noindent
 \textbf{Theorem 2}. (See \cite{GP-T2}) 
 \emph{The sequence $(N,\Sigma^{(j)},\mathbf{\Gamma}^{(j)})$ stabilizes for any order that 
 is lexicographic with respect to some basis of $N$. 
 In particular it stabilizes for the "divisorial" preorders associated to vectors of $N$.}


\end{talk}


\end{report}


\end{document}

